Author's response to reviews

Title: Prevalence and factors associated with Posttraumatic Stress Disorder seven years after the conflict in three districts in northern Uganda (The Wayo-Nero Study)

Authors:

James Mugisha (jmmugi77@hotmail.com)
Herbert Muyinda (muyindaherbert@hotmail.com)
Peter Wandiembe (symon.wand@gmail.com)
Eugene Kinyanda (Eugene.Kinyanda@mrcuganda.org)

Version: 6  Date: 13 June 2015

Author's response to reviews:

see letter attached